Author's response to reviews

Title: Management of renal extraskeletal mesenchymal chondrosarcoma

Authors:

Vitalie Gherman (vitaliegherman@gmail.com)
Ciprian Tomuleasa (ciprian.tomuleasa@umfcluj.ro)
Catalina Bungardean (mariabun2002@yahoo.com)
Dan Vasile Stanca (vasilestanca@yahoo.com)
Victor Dan Ona (ona_victor@yahoo.de)
Bogdan Feciche (vitaliegherman@yahoo.com)
Alexandru Irimie (office@iocn.ro)
Ioan Coman (jcoman@yahoo.com)

Version: 8
Date: 31 August 2014

Author's response to reviews: see over
To the editorial board,

Thank you for your reply and the recommendations you have made.

As recommended by Dr. Jens Bedke in his second revision of our manuscript, we have removed two images (Figure 3 and 4).

As a consequence, we have submitted a revised manuscript and a revised image legend.

Regarding the documentary evidence of the consent to publish obtained from the patient, we have emailed you the attachment, and did not include it as a part of submission, as you have advised.

If you will need further information, please do not hesitate to contact us.

Sincerely yours,

Vitalie Gherman, M.D.

Department of Urological and Robotic Surgery, Clinical Municipal Hospital,

Cluj Napoca, Romania

E-mail: vitaliegherman@gmail.com